Author's response to reviews

Title: A functional polymorphism in the SPINK5 gene is associated with asthma in a Chinese Han Population

Authors:

Qiji Liu (liuqiji@sdu.edu.cn)
Yu Xia (xiaxixi716@yahoo.com.cn)
Wenjing Zhang (zwiviolin@yahoo.com.cn)
Jisheng Li (lijisheng@gmail.com)
Pin Wang (wangpin@mail.sdu.edu.cn)
Huaichen Li (lihuaichen@yahoo.com.cn)
Chunhua Wei (weichunhua@163.com)
Yaoqin Gong (yxg8@sdu.edu.cn)

Version: 6 Date: 15 June 2009

Author's response to reviews: see over
Dear editor,

Thanks for accepting our manuscript. According to the suggested revision, we have made a careful check and change including authors’ contribution, table format and figure title. Also, each word was checked to make sure no typographical errors. Many thanks for your help.

Best regards

Qiji Liu
Institute of Medical Genetics,
School of Medicine, Shandong University
Jinan 250012
P.R.China
Tel:86-531-88382043
Fax:86-531-88382115
E-mail: liuqiji@sdu.edu.cn